Author’s response to reviews

Title: Cross-sectional area of the paraspinal muscles and its association with muscle strength among fighter pilots: A 5-year follow-up

Authors:

Tuomas Honkanen (tuomas.honkanen@mil.fi; honkanenjt@gmail.com)
Matti Mäntysaari (matti.mantysaari@mil.fi)
Tuomo Leino (tuomo.leino@mil.fi)
Janne Avela (janne.avela@jyu.fi)
Liisa Kerttula (liisa.kerttula@hus.fi)
Ville Haapamäki (ville.haapamaki@hus.fi)
Heikki Kyröläinen (heikki.kyrolainen@jyu.fi)

Version: 1 Date: 16 Nov 2017

Author’s response to reviews:

Here are the responses for editor comments:

There is now statement in the manuscript that participant consent was written (not verbal). The statement has been amended in both method section and to the declarations.

Furthermore, a written informed consent for the publication of figures and information of data and material availability is provided in the declarations of the revised version.

The figure titles have been moved to under the separate heading of "Figure Legends" in the revised version.